Author's response to reviews

Title: Periodontal disease and atherosclerosis from the viewpoint of the relationship between community periodontal index of treatment needs and brachial-ankle pulse wave velocity.

Authors:

Koichi Miyaki (miyaki@sc.itc.keio.ac.jp)
Katsunori Masaki (masaki@keiommed.net)
Mariko Naito (mnaito@med.nagoya-u.ac.jp)
Toru Naito (naito@college.fdcnet.ac.jp)
Keika Hoshi (keikah@jcqhc.or.jp)
Shugo Tohyama (shugotohyama@w4.dion.ne.jp)
Asako Hara (Asako2307@aol.com)
Takeo Nakayama (nakayama@pbh.med.kyoto-u.ac.jp)

Version: 5 Date: 10 April 2006

Author's response to reviews:

Dear Editor,

Thank you very much for your e-mail dated May 29, 2006 that favoured our manuscript. We have revised it according to the following comments. We would greatly appreciate it if you could kindly find it acceptable.

Sincerely yours,

Takeo Nakayama

Many thanks for returning the revised version of your manuscript (above) in response to the referees' comments. I am delighted to say we consider the referees' criticisms satisfactorily addressed, and anticipate acceptance of your revised manuscript.

However, before acceptance we feel that some improvements are needed in the use of language in the manuscript. We recommend that you ask a native English speaking colleague to help you copyedit the paper. If this is not possible, please make use of a professional copyediting service. We suggest the 'Manuscript Presentation Service' based at the University of Aberdeen. Details and contact information are available from the following link http://www.abdn.ac.uk/mps/. There are also suggestions for copyediting services in the 'Style and Language' section of our instructions to authors at http://www.biomedcentral.com/info/authors/. Please indicate in the covering letter that will accompany your resubmission how the language corrections have been attended to.

We would also request that you go through the manuscript formatting checklist one more time and ensure that your revised manuscript conforms to all of the points. The link to the formatting checklist is provided at the bottom of this e-mail. Please could you also make the following changes:

Title- Please remove the heading 'Title', please also remove all unnecessary capitalization from the title, and remember to update the details on the online submission page accordingly.

Authors- Please remove the heading 'Authors'.
Affiliations- Please remove the heading 'Institutions'. Please use superscript numbers to link each author to their institution.

Keywords- Please remove the keywords section.

Authors' contributions- Please use initials to refer to each author.

Conclusions- Please include a brief Conclusions section after the Discussion.

References- Please list all of the authors in the references and avoid the use of the suffix et al.

Tables- Please format the tables in black and white.

When you have made these specific changes and any other necessary changes to the format, please upload the revised version by following the instructions provided below.

To submit your revised manuscript
-----------------------------------

1. Go to 'my BioMed Central' (http://www.biomedcentral.com/my/) and log on with your email address and password. Then click on 'My manuscripts'. You will reach a page giving details of all the manuscripts you have submitted to journals published by BioMed Central.

2. Click on the 'Submit revision' button next to the title of this manuscript to go through to a submission form.

3. With the 'Manuscript details' tab, please update the title, abstract and author details if they have changed since the previous version. It is very important that all changes are updated on this page, as well as in the manuscript file as the information on this page will be used in PubMed and on BioMed Central.

4. With the 'Cover letter' tab, please provide a covering letter with a point-by-point description of the changes made.

5. With the 'Upload files' tab, please upload the revised version of the manuscript and press 'Submit new version'. Please wait for the confirmation page to appear - this may take a few moments.

Please do not hesitate to contact us if you have any problems or questions.

With best wishes,

Iratxe

Iratxe Puebla, Senior Assistant Editor
BMC-series journals

email: editorial@biomedcentral.com
Web: www.biomedcentral.com

Manuscript formatting checklist- http://www.biomedcentral.com/info/ifora/medicine_journals

This email has been scanned by Postini.
For more information please visit http://www.postini.com